Author's response to reviews

Title: Hypothyroidism among military infants born in countries of varied iodine nutrition status

Authors:

Marcus M. Cranston (marcus.cranston@med.navy.mil)
Margaret A.K. Ryan (margaret.ryan@med.navy.mil)
Tyler C. Smith (tyler.c.smith@med.navy.mil)
Carter J. Sevick (carter.sevick@med.navy.mil)
Stephanie K. Brodine (stephanie.brodine@med.navy.mil)

Version: 6 Date: 29 January 2010

Author's response to reviews: see over
There was a slight revision to two manuscript numbers in the second sentence of the second paragraph of the Discussion.
Many Thanks!